Author’s response to reviews

Title: Open globe and penetrating eyelid injuries from fish hooks

Authors:

Konstantine Purtskhvanidze (konstantine.purtskhvanidze@uksh.de)
Mark Saeger (mark.saeger@uksh.de)
Felix Treumer (felix.treumer@uksh.de)
Bernhard Nölle (bernhard.noelle@uksh.de)
Johann Roider (johann.roider@uksh.de)

Version: 3 Date: 13 Jan 2019

Author’s response to reviews:

Author’s response to reviews

Title: Open globe and penetrating eyelid injuries from fish hooks

Authors:

Konstantine Purtskhvanidze (konstantine.purtskhvanidze@uksh.de)
Mark Saeger (mark.saeger@uksh.de)
Felix Treumer (felix.treumer@uksh.de)
Bernhard Nölle (bernhard.noelle@uksh.de)
Johann Roider (johann.roider@uksh.de)

1. Please amend the ages presented in your manuscript so they appear as age ranges (e.g as <18 or >=18).

We amended the ages and therefore also deleted the patients’ number (table 1 line 129, lines 108, 141, 143, 150, 152, 166, 167, 168, 293, 294, 298, 303, 309).

2. Please also amend the statement included in the Consent for publication section so that it clearly states that you have obtained written consent to publication from the 3 patients (or their parent/legal guardian) for the images and details included in the figures.

We included this statement (line 235).

3. Please also remove the competing interests information that is currently in the acknowledgements section. If you have no further acknowledgements please put “Not Applicable” in the Acknowledgements section.

We included this statement (line 220, 250).